Author’s response to reviews

Title: Predictors of poor nutritional status among children aged 6-24 months in agricultural regions of Mali: a cross-sectional study

Authors:
Caroline Makamto Sobgui (caroline.sobgui@worldveg.org)
Leopold Fezeu Kamedjie (l.fezeu@eren.smbh.univ-paris13.fr)
Fatou Diawara (diawarafatou@gmail.com)
Honafing Diarra (honafing.diarra@worldveg.org)
Victor Afari-Sefa (victor.afari-sefa@worldveg.org)
Abdou Tenkouano (a.tenkouano@coraf.org)

Version: 1 Date: 29 Mar 2017

Author’s response to reviews:

Makamto Sobgui Caroline
World Vegetable Centre
West and Central Africa office
P.o.Box 320
Bamako, Mali
0022393646426
cmsobgui@yahoo.com
carolinesobgui@worldveg.org
Bamako, 29th May 2017

Dear Madam /Sir,

I wish to resubmit the manuscript entitled Predictors of poor nutritional status among children aged 6 – 24 months in agricultural regions, Mali” for consideration by the BMC Nutrition Journal.

I confirm based on the comment made by the editor’s comments that the previous manuscript has a high similarity check. I have rewritten the Background and the Discussion sections to make sure that there is not overlap with previous or existing literature. I also revised the Reference section.

Thank you for your consideration of this manuscript.

Sincerely,

Caroline Makamto Sobgui